Author’s response to reviews

Title: The role of T2* Gradient Echo in the diagnosis of tumefactive intrahepatic extramedullary hematopoiesis in myelodysplastic syndrome and diffuse hepatic iron-overload: A case report and literature review.

Authors:

abel belay (abelassefamd@gmail.com)
Andrew Bellizzi (andrew-bellizzi@uiowa.edu)
Alan Stolpen (alan-stolpen@uiowa.edu)

Version: 5 Date: 28 Nov 2017

Author’s response to reviews:

Give detailed physical and neurological examination on his first admission.
Physical and neurological exam was normal, and included in the case report
What was the temperature, pulse, blood pressure and temperature, on admission?
Vitals were normal, included in the case report
All abbreviations need to be explained on the first time they are used.
Abbreviations have been explained.
Give antibiotics given before and after surgery with duration and dose.
No antibiotics were given
Give all results of more laboratory findings (i.e. CBC, liver and renal functions), urinalysis, serology, microbiology etc)
Relevant lab findings were included in the case report

Was an autopsy performed, and what were the findings?
No autopsy performed; statement saying so included in the case report
In the Discussion – Start with summarizing the case and describe what is unique in this case compared to what is available in the literature.

Summary of the case report and what makes it unique included at the beginning under the discussion